

G OPEN ACCESS

Citation: Pereira SH, Sá Magalhães Serafim M, Moraes TdFS, Zini N, Abrahão JS, Nogueira ML, et al. (2024) Design, development, and validation of multi-epitope proteins for serological diagnosis of Zika virus infections and discrimination from dengue virus seropositivity. PLoS Negl Trop Dis 18(4): e0012100. https://doi.org/10.1371/journal. pntd.0012100

Editor: Michael W. Gaunt, Solena Ag, UNITED STATES MINOR OUTLYING ISLANDS

Received: November 24, 2023

Accepted: March 22, 2024

Published: April 18, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0012100

Copyright: © 2024 Pereira et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data included in the manuscript is freely available by accessing

RESEARCH ARTICLE

Design, development, and validation of multiepitope proteins for serological diagnosis of Zika virus infections and discrimination from dengue virus seropositivity

Samille Henriques Pereira¹, Mateus Sá Magalhães Serafim², Thaís de Fátima Silva Moraes¹, Nathalia Zini^{3,4}, Jônatas Santos Abrahão², Maurício Lacerda Nogueira³, Jordana Grazziela Alves Coelho dos Reis¹, Flávia Fonseca Bagno⁴, Flávio Guimarães da Fonseca^{1,4}*

 Laboratório de Virologia Básica e Aplicada, Departamento de Microbiologia, Universidade Federal de Minas Gerais, Belo Horizonte, Minas Gerais, Brazil, 2 Laboratório de Vírus, Departamento de Microbiologia, Universidade Federal de Minas Gerais, Belo Horizonte, Minas Gerais, Brazil, 3 Laboratório de Pesquisa em Virologia, Faculdade de Medicina de São José do Rio Preto, São José do Rio Preto, São Paulo, Brazil,
Centro de Tecnologia em Vacinas, Universidade Federal de Minas Gerais, Belo Horizonte, Minas Gerais, Brazil

* fdafonseca@icb.ufmg.br

Abstract

Zika virus (ZIKV), an arbovirus from the Flaviviridae family, is the causative agent of Zika fever, a mild and frequent oligosymptomatic disease in humans. Nonetheless, on rare occasions, ZIKV infection can be associated with Guillain-Barré Syndrome (GBS), and severe congenital complications, such as microcephaly. The oligosymptomatic disease, however, presents symptoms that are quite similar to those observed in infections caused by other frequent co-circulating arboviruses, including dengue virus (DENV). Moreover, the antigenic similarity between ZIKV and DENV, and even with other members of the *Flaviviridae* family, complicates serological testing due to the high cross-reactivity of antibodies. Here, we designed, produced in a prokaryotic expression system, and purified three multiepitope proteins (ZIKV-1, ZIKV-2, and ZIKV-3) for differential diagnosis of Zika. The proteins were evaluated as antigens in ELISA tests for the detection of anti-ZIKV IgG using ZIKV- and DENVpositive human sera. The recombinant proteins were able to bind and detect anti-ZIKV antibodies without cross-reactivity with DENV-positive sera and showed no reactivity with Chikungunya virus (CHIKV)- positive sera. ZIKV-1, ZIKV-2, and ZIKV-3 proteins presented 81.6%, 95%, and 66% sensitivity and 97%, 96%, and 84% specificity, respectively. Our results demonstrate the potential of the designed and expressed antigens in the development of specific diagnostic tests for the detection of IgG antibodies against ZIKV, especially in regions with the circulation of multiple arboviruses.

https://github.com/SamilleHenriques/Paper-ZIKVmultiepitope.

Funding: FGF received funds from Fundação de Amparo à Pesquisa do Estado de Minas Gerais -FAPEMIG (grant numbers APQ-03081-17, APQ-04295-17 and RED-00282-16); and from Conselho Nacional de Desenvolvimento Científico e Tecnológico - CNPq (grant # 402256/2022-0). The study was also financed by the Coordenação de Aperfeiçoamento de Pessoal de Nível Superior – Brasil (CAPES) – Finance Code 001, through fellowships given to post-graduation students involved in the work. FGF, MLN, JGACR and JSA are CNPQ-PQ fellowship recipients. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

From 2015 onward, Zika virus (ZKV) caused major epidemics in Brazil and other South American countries, where many children were born with congenital problems, such as microcephaly, as a result of infections in pregnant women. Additionally, many cases of Guillain Barré Syndrome in adults were also caused by infections with the virus. Detection of ZKV infection is, therefore, very important; nonetheless, most patients infected by ZKV develop either no symptoms or symptoms that are not distinguishable from diseases caused by other co-circulating arboviruses, including dengue. To complicate matters, Zika, and dengue viruses show intense genetic resemblance, and serological diagnosis presents high levels of uncertainty due to extensive cross-reactivity. In this work, we developed proteins to serologically differentiate Zika from dengue infections with no or minimum cross-reactivity. We produced three multiepitope, recombinant proteins which were able to differentiate zika-positive samples from dengue-positive sera. The tests employing the recombinant proteins had high sensitivity and specificity, especially in the case of ZIKV-1 and ZIKV-2 proteins, whilst the ZIKV-3 protein was less efficient. The recombinant proteins showed clear potential to be used as diagnostic tools for detecting antibodies against ZIKV infections with no cross-reactivity to other prevalent arbovirus.

Introduction

The Orthoflavivirus zikaense (ZIKV) is an arthropod-borne virus (arbovirus) transmitted by *Aedes* mosquitoes. It belongs to the *Flaviviridae* family, together with other viruses of public health importance, such as *Orthoflavivirus flavi* (YFV), *Orthoflavivirus denguei* (DENV) and *Orthoflavivirus nilense* (WNV) [1,2]. The emergence of ZIKV in the Americas in 2015 brought great concern, as infections with this virus were associated with serious complications, such as congenital Zika syndrome (CZS). During the epidemics in Brazil, more than 2,630 cases of microcephaly linked to ZIKV infections were registered, in addition to other congenital complications and birth defects, as well as GBS in adults [3]. Nonetheless, most ZIKV infections in humans are asymptomatic, making it difficult to assess the real size of the epidemics. Estimates have put the number of cases in the range of 1.5 million, in Brazil, during the two years of epidemic peak [4]. Although the emergency period ended in July 2017, new cases of CZS and deaths associated with the disease have been reported in low frequencies in the Country. Until the end of 2022, 9,204 probable cases were reported, corresponding to an incidence rate of 4.3 cases per 100,000 Brazilian inhabitants [5,6].

Common clinical manifestations of Zika include fever, rash, headaches, joint pain, conjunctivitis, and muscle pain [7]. The most frequent symptoms are similar to those of other arbovirus diseases, such as dengue and chikungunya, which make clinical diagnosis difficult [8]. The World Health Organization (WHO) recommends the diagnosis by reverse transcriptase quantitative polymerase chain reaction (RT-qPCR), but its use is limited due to the narrow detection window of viral RNA in body fluids [9,10]. Serological tests available on the market, on the other hand, are complicated by high cross-reactivity between different flaviviruses, due to the high genomic and antigenic similarities among them [10]. Cross-reactivity is a problematic issue in places where ZIKV and DENV co-circulate, as is the case in the Americas [11,12]. In this scenario, Plaque Reduction Neutralization Tests (PRNT), which measure specific neutralizing antibodies, could be an alternative; however, the technique is quite laborious and may be imprecise when assessing samples of patients with previous DENV infections, as well as those in the early days after infection [13,14]. As a result, ZIKV diagnosis is not precise, resulting in under-reporting and misdiagnosing (e.g., DENV). Indeed, our group has previously detected many misdiagnosed or not reported cases of Zika, Dengue, or Chikungunya from suspected patient samples collected during 2015–2017 and sent to the reference state laboratory for diagnosis [15].

The non-structural 1 (NS1) glycoprotein, a protein secreted in large amounts during the viremic phase of flavivirus infections, has been widely used for the development of immunoassays. However, the protein is highly conserved in flaviviruses, presenting up to 62.5% of identity among them [4,16]. Although the NS1 and E proteins are the most immunogenic proteins, all viral proteins have antigenic determinants that could be exploited for precise serological diagnosis [17]. Thus, the use of chimeric proteins with the association of several epitopes that do not show cross-reactivity can increase the performance of immunoassays.

Considering the relevance of ZIKV diagnostics and the need to develop specific and sensitive serological assays for its correct diagnosis, we developed a number of multi-epitope recombinant proteins to be used as antigens and evaluated their specificity and sensitivity in ELISA tests.

Methods

Human sera samples and Ethics statement

Sera samples from ten confirmedly positive zika-infected patients (herein termed ZIKV+), ten positive samples for dengue (DENV+), eight positive samples for Chikungunya (CHIKV), and eight samples with elevated levels of rheumatoid factor (RF) were provided by the Laboratory of Virology and Rickettsioses of the Ezequiel Dias Foundation (FUNED). These sera were characterized by commercial ZIKV and DENV IgG kits or RT-PCR, and ultimately characterized by PRNT. Additionally, a second set of sera composed of fifty ZIKV+/DENV- samples, twenty-six DENV+/ZIKV- samples, and fifty-one negative samples for Zika and dengue (ZIKV-/DENV-) were provided by the Virology Research Laboratory of the São José do Rio Preto Medical School (FAMERP). These sera were obtained from a cohort study and characterized by commercial ZIKV and DENV IgG kits, and by PRNT.

Written authorizations to use the sera bank were provided by both Institutions' research boards (Ezequiel Dias Foundation–FUNED, and São José do Rio Preto Medical School—FAMERP). Sera samples from FAMERP were collected upon formal consent from each participant. Sera from FUNED were intended for infection diagnosis and were protected by patient anonymity.

Multi-epitope recombinant proteins

For the construction of multi-epitope proteins, B cell linear epitopes previously described in the literature and that had been tested in serological assays were annotated [18–21]. From this dataset, epitopes with the lowest percentage of aminoacidic identity between ZIKV and DENV were selected. At the end, seven epitopes were selected, three of them described by Lee et al. (2018) [18], and four epitopes described by Kam et al. (2018) [20]. Three multi-epitope proteins were designed and the epitopes were connected by glycine and proline linkers (GPGPG) as previously described [22]. The amino acid sequences of each final multiepitope polypeptide are shown in Table 1. Nucleotide sequences were commercially synthesized and inserted into pET21a+ plasmid (Genone-Biotech, Rio de Janeiro, Brazil) with the addition of the coding sequence for six histidine residues at the COO- end of each protein.

Modelling and conformational analysis of multi-epitope proteins

Because these were artificially designed proteins, we modeled each one beforehand in order to assess their potential stoichiometric viability. Therefore, the multi-epitope amino acid

ZIKV-1	ZIKV-2	ZIKV-3	
HKKGEARRSRRAVTLPSHG	HMCDATMSYECPMLDEGV	HMCDATMSYECPMLDEGV	
PGPGTGVFVYNDVEAWRDRYKY	GPGPGHKKGEARRSRRAV	GPGPGHKKGEARRSRRAVT	
GPGPGWGKSYFVRAAKTNNSFVVDGDT	TLPSHGPGPGTGVFVYND	LPSHGPGPG TVNMAEVRSYC	
LKECPLKHGPGPGPAVIGTAVKGKEAVH	VEAWRDRYKYGPGPGWGKS	YEASIS GPGPGTGVFVYNDVE	
	YFVRAAKTNNSFVVDGDTLKECP	AWRDRYKYGPGPG SVEGELNA	
	LKHGPGPGPAVIGTAVKGKEAVH	ILEENGV GPGPGWGKSYFVRA	
		AKTNNSFVVDGDTLKECPLKH	
		GPGPGPAVIGTAVKGKEAVH	

Table 1. FASTA sequences of multi-epitope proteins. Highlighted are the additional epitopes to each sequence.

https://doi.org/10.1371/journal.pntd.0012100.t001

sequences (FASTA) of ZIKV-1, ZIKV-2, and ZIKV-3 were submitted to the AlphaFold software (version 2.3.1) (https://github.com/sokrypton/ColabFold) [23,24] for prediction of their three-dimensional (3D) protein structures with atomic accuracy. Modeled structures were aligned and analyzed with the PyMOL software (v0.99c) [25]. Figures were also generated with PyMOL.

Identification of potential binding sites and pockets of multi-epitope proteins

To check whether the potential binding residues and regions were correctly exposed on the AlphaFold2 modeled proteins' surface, we employed the FTSite [26] and FTMap [27] webservers. These modeling tools predict the presence of potential protein cavities that may act as binding sites or hot spots, that is, relevant residues usually responsible for interactions (e.g., protein-protein interactions), such as antigen-antibody ligations. Molecule clusters from the FTMap in each predicted site can also suggest the affinity of these potential binding regions.

Solvent accessible surface areas (SASA) analyses

The calculation of solvent accessible surface areas (SASA) was assessed with the GETAREA software [28] (https://curie.utmb.edu/getarea.html). Modeled structures were submitted to the webserver in PDB format, resulting in polar, non-polar, and total area/energy of each protein, including the total number of accessible surface atoms and assessment of buried, not accessible atoms. Parameters were set as default for the radius of the water probe (A) = 1.4 with no gradient included in calculations.

Expression and purification of recombinant proteins

Competent *Escherichia coli* BL21 (DE3) were transformed with the synthesized plasmids and induced for 3 h with 1 mM IPTG (Invitrogen, Massachusetts, USA). Proteins were purified by affinity chromatography using a His-Trap HP column (GE Healthcare, Chicago, Illinois, USA) within the Äkta Start purification system (GE Healthcare Life Sciences, Chicago, Illinois, USA). The proteins were separated by SDS-PAGE and transferred onto nitrocellulose membranes (Cytiva, Danaher Corporation, USA) for western blot evaluation. Subsequently, the membranes were incubated with 1:1000 anti-histidine (Invitrogen, Massachusetts, USA) for 2 h. As a secondary antibody, anti-mouse IgG 1:40,000 (Sigma–Aldrich, San Luis, Missouri, USA) was used. Similarly obtained blots were also incubated with a sera pool from ZIKV + patients (10 sera samples) for 2 h and incubated for another hour with anti-human IgG 1:25,000 (Sigma–Aldrich, San Luis, Missouri, USA). The detection was done using Clarity Western ECL Substrate (Bio-rad, Hercules, California, USA) followed by visualization with image software ChemicDoc (Bio-Rad, Hercules, California, USA).

In-house indirect ELISA anti-ZIKV (IgG)

The 96-well microplates (High Binding, Sarsterd, Germany, Cat. 82.1581.200) were coated with 400 ng/well of recombinant proteins and incubated overnight at 4°C. Subsequently, microplates were blocked for 2 h at 37°C with skim powder milk 5% (Molico-Nestlé, Vevey, Switzerland). Plates were incubated with sera samples ZIKV+, DENV+, and ZIKV-/DENV-(1:100, Na₂HPO₄ 100 mM, KH₂PO₄ 18 mM, NaCl 1500 mM, KCl 28 mM, sucrose 2%, Tween 20 1%, Tween 80 0,1%, Proclin 0,1%, skim milk powder 5%) for 30 minutes at 37°C followed by two standard washes, a single wash with urea [6M] solution, and two more standard washes, as previously described [29]. Secondary anti-human IgG antibodies labeled with peroxidase 1:70,000 (Sigma-Aldrich, San Luis, Missouri, USA) were added for 30 minutes at 37°C. Results were revealed with TMB (Scienco Biotech, Santa Catarina, Brazil). Absorbances were measured at 450 nm in a GO Microplate Spectrophotometer (Thermo Fisher Scientific, Waltham, Massachusetts, USA). The cut-off point was defined by the sum of the mean optical density (OD) of the negative controls plus the standard deviation. The indexes were calculated by the OD of each sample divided by the cut-off value. Results above 1.1 were considered positive; those below 0.8 as negative, and values between 0.8 and 1.1 were classified as borderline results. Positivity is given by the ratio of positive reacting samples over the total of true positive Zika samples. Similarly, sensitivity is given by the ratio of positive reacting samples over true positive samples plus false negative results, removing borderline results from sample totals. Lastly, specificity is given by the ratio of negative reacting samples over the total of true negative samples plus false positive results, also removing borderline results from sample totals.

Statistical analyses

Statistical analyses were performed using GraphPad Prism 9 program with the Two-way ANOVA test with multiple comparisons. Non-parametric statistics provide an alternative approach to significance testing which could be explored in future research. A receiver operating characteristic (ROC) curve, was applied to evaluate the ELISA test performance as a statistical tool for the global accuracy of the test.

Results

Multi-epitope protein modeling, conformational and potential binding sites' analyses, and accessible surface calculations of ZIKV-1, ZIKV-2 e ZIKV-3

Initially, ZIKV multi-epitope protein models were obtained with AlphaFold, and the predicted structures were compared in a 3D alignment, showing their structural differences. ZIKV-1 (Fig 1A) presents a shorter amino acid chain when compared to ZIKV-2 (Fig 1B), which presents a **HMCDATMSYECPMLDEGVGPGPG** polypeptide addition (position 1 to 23). Lastly, both are shorter than ZIKV-3 (Fig 1C), which also presents a **HMCDATMSYECPML-DEGVGPGPG** addition (position 1 to 23), as well as two intercalations

GPGPGTVNMAEVRSYCYEASIS (position 42 to 63) and **GPGPGSVEGELNAILEENGV** (positions 87 to 106). Interestingly, their 3D alignment (Fig 1D), showed similar spatial occupancy in all three structures, despite their differences in size, but also highlighted their conformational and folding variations, implied from the sequences' additions for ZIKV-2 and ZIKV-3, and both sequences' intercalations within ZIKV-3.

Subsequently, to compare the presence of their respective, potential binding sites, predictions employing FTSite and FTMap softwares were analyzed. First, the three predicted binding sites within the ZIKV-1 structure were shown to be in tight, close formation, according to the



Fig 1. Models of the multi-epitope proteins. Modeled structures obtained with AlphaFold are represented by cartoons with transparent surfaces; (A) ZIKV-1 (pale cyan). (B) ZIKV-2 (aquamarine) presents a sequence addition (gray), as well as (C) ZIKV-3 (light teal), which also presents two sequences intercalation (ruby), evidencing structures' differences in size. (D) 3D alignment corroborates their structural differences, also showing a similar spatial occupation among the three proteins and their folding variations. Images were generated with PyMOL (v0.99c).

https://doi.org/10.1371/journal.pntd.0012100.g001

FTSite (Fig 2A). This was slightly different from the structural prediction in ZIKV-2, showing a slight move of site #1 (salmon) and a shift of site #2 (lime green) towards the added sequence in position 1 to 23 (Fig 2B). Interestingly, ZIKV-3 potential sites were closely related to ZIKV-1, but with a deviation towards the two sequences intercalations (Fig 2C), with both sites being mostly covered by this part of the structure, which could result in a lower sensitivity. In addition, the FTSite results were also corroborated by the probes' clustering predicted with the FTMap (Fig 5D–5F), corroborating their preferable regions in which bindings would occur, such as protein-protein interactions.

Sequence additions (ZIKV-2 and ZIKV-3) and intercalations (ZIKV-3) resulted in conformational and folding changes in the two last predicted structures. Especially for ZIKV-3, which has the longer amino acid sequence, suggesting structural changes or additional protein-protein interactions. To evaluate these potential differences in proteins' accessibility, we also calculated the solvent accessible surface area (SASA) of the three multi-epitope proteins, which showed an evident increase in the number of available surface atoms and total area/ energy of interaction for ZIKV-3 (Table 2). Taken together, these would potentially result in more unspecific protein-protein interactions from the sequence additions and/or intercalations, thus supporting the lower specificity observed in the empiric tests. Lastly, smaller differences in SASA calculations between ZIKV-1 and ZIKV-2 could also support their slight differences as observed in the empiric testing.



Fig 2. Potential binding sites and probes clustering predicted in the multi-epitope proteins. Three potential binding sites were predicted by FTSite: #1 (salmon), #2 (lime green), and #3 (blue), all represented by a tridimensional mesh in (A) ZIKV-1 (pale cyan), (B) ZIKV-2 and (C) ZIKV-3. Probes clustering predicted by FTMap for (D) ZIKV-1, (E) ZIKV-2, and (F) ZIKV-3, corroborate FTSite predictions. All images were generated using PyMOL software (v0.99c).

https://doi.org/10.1371/journal.pntd.0012100.g002

Expression and purification of recombinant proteins

ZIKV multi-epitope proteins were produced in transformed *E. coli* and the production was confirmed by SDS-PAGE. Proteins were purified using nickel affinity chromatography (HisTrap) and western blot analysis with a pooled sera sample from ZIKV+ patients (Fig 3).

Development of an in-house anti-ZIKV IgG ELISA and evaluation of the recombinant multi-epitope proteins as antigens

The multi-epitope proteins produced were evaluated as antigens in an indirect ELISA assay. For the tests, 60 ZIKV+ sera samples; 51 double negative samples (negative for ZIKV and DENV); 36 DENV+ samples; eight CHIKV+ samples; and eight RF+ samples were used. The ZIKV-1 protein (Fig 4A) reacted with 52 of the 60 ZIKV+ samples tested (86.6% positivity), with three samples classified as indeterminate results (91.2% sensitivity); and one out of the 51 negative samples showed a false-positive result (98.1% specificity). In addition, regarding DENV+ samples, only three out of 36 samples showed some reaction with ZIKV-1, and five were placed in the grey zone (90.3% specificity). Lastly, in relation to the eight CHIKV+ and RF+ samples, ZIKV-1 presented no reaction, suggesting 100% specificity. The ZIKV-2 protein (Fig 4B) reacted with 57 out of the 60 ZIKV+ samples with no borderline results (95% positivity and sensitivity), and two negative samples showed false-positive results, also with no indeterminate results (96.2% specificity). Regarding cross-reactivity, it presented 100% specificity against 36 DENV+ samples, with one in the grey zone, and 100% specificity against the CHIKV- and RF-positive sera samples. Lastly, the ZIKV-3 (Fig 4C) protein showed the lowest performance in comparison to the first two multi-epitope proteins, reacting with 32 out of the 60 ZIKV+ sera samples (53.3% positivity), and 12 samples in the grey zone (66.6% sensitivity). In addition, it also showed lower specificity (91%) regarding DENV+ samples, with nine out of 36 samples tested being in the grey zone. As for the CHIKV+ and RF+ samples, tested as possible interferents, ZIKV-3 presented 100% specificity, although one CHIKV+ sera was placed in the grey zone after testing.

Parameters	ZIKV-1	ZIKV-2	ZIKV-3
Polar area/energy	3769.77	4131.56	5207.06
Apolar area/energy	8151.01	8799.85	10512.21
Total area/energy	11920.78	12931.41	15719.28
Number of buried/ unavailable atoms	86	124	168
Number of surface accessible atoms	643	767	1016

https://doi.org/10.1371/journal.pntd.0012100.t002

To confirm the obtained results, the developed ELISA was independently tested at the Virology Research Laboratory (at FAMERP, SP, Brazil) accessing a second panel of sera samples. For this independent evaluation, 36 ZIKV+ samples, 96 double negative samples (negative for ZIKV and DENV), and 96 DENV+ samples were used. ZIKV-1 protein (Fig 5A) reacted with 32 out of the 36 ZIKV+ samples (88.9% positivity), with one sample in the grey zone, and one sample as a false-negative result (88.9% sensitivity). As for the negative samples, three out of the 96 tested presented false-positive results and three were in the grey zone (96.9% specificity). Regarding DENV+ samples as interferents, four samples out of the 96 tested reacted positively (95% specificity). The ZIKV-2 protein (Fig 5B) reacted with 35 out of the 36 samples tested, with no borderline results (97.2% positivity and sensitivity). Among the 96 negative samples tested, three showed false-positive results (96.9% specificity). Regarding DENV+ samples as interferents, ZIKV-2 reacted with only one sample, and one was in the grey zone (98.9% specificity). Similar to the first set of samples, the ZIKV-3 protein (Fig 5C) showed the lowest performance, reacting with 19 out of 36 ZIKV+ samples (52.8% positivity), with 11 in the grey zone (76% sensitivity). Of the 96 negative samples tested, it reacted with one sample, and 17 were in the grey zone (82% specificity). Regarding the 96 DENV+ samples, four showed a false-positive result (94.9% specificity), but a considerable number of samples were in the grey zone (n = 17).

Discussion

The diagnosis of ZIKV is challenging, as most infected humans do not present readily discernible clinical signs [30]. Furthermore, clinical diagnosis in endemic areas is difficult due to symptoms similarities with other arboviral diseases [31]. To further complicate things, high genetic and antigenic similarities between circulating flaviviruses, especially ZIKV and DENV [11,32] make serological diagnosis imprecise. Currently, Brazil is facing the simultaneous circulation of Dengue, Zika and Chikungunya viruses, with a severe epidemic of Dengue leading to a high number of cases and deaths. Probable cases of Zika are currently low, considering the historical series; however, as of February 2024, 13 probable cases in pregnant women have been reported, two of which have been confirmed [33], showing the need for effective tests in the public health system. Nonetheless, it is worth noting that the system may be overloaded due to the dengue outbreak and therefore it is possible that not all samples are being tested for all arboviruses, and Zika cases may be underreported. Therefore, the search for antigens that allow specific detection of antibodies directed to these viruses is challenging [11,34,35]. Although ZIKV infection is self-limiting, it is important to follow up and monitor cases due to their association with GBS in adults, sequelae in pregnant women (e.g., GBS complications), neurological newborn disorders, congenital syndrome, birth defects, and even perinatal death or stillbirth [35-38].

Regarding diagnosis, full-length, recombinant NS1 is commonly used in commercial anti-ZIKV and anti-DENV ELISA kits. Diagnostic tests also use the E protein, either full-length or



Fig 3. Production of ZIKV-1, ZIKV-2, and ZIKV-3 multi-epitope proteins. (A) SDS-PAGE of cell extracts after induction with 1mM IPTG showing the expression of proteins ZIKV-1 (14 kDa), ZIKV-2 (18 kDa), and ZIKV-3 (20 kDa). (B) Multi-epitope proteins after purification by nickel affinity chromatography using HisTrap columns. (C) Western blot of multi-epitope proteins with pooled sera from ZIKV+ patients. M: molecular weight standards; CN: non-transformed cells; NI: transformed and non-induced cells; I: cells induced with 1M IPTG; ZIKV-E: ZIKV envelope protein (recombinant, positive control).

https://doi.org/10.1371/journal.pntd.0012100.g003

a truncated form in which the transmembrane region is removed to facilitate the recombinant protein production [39]. ZIKV and DENV have about 60% amino acid sequence identity, on average, with E and NS1 proteins having 61.4% and 62.5% identity, respectively [17]. The high homology of these proteins impacts the specificity of available tests, challenging the correct diagnosis of Zika and dengue, especially in endemic areas [40]. Therefore, the development of serological tests based on chimeric recombinant proteins capable of differentiating Zika-positive samples from dengue-positive sera is a feasible option to help with differential diagnosis, even in highly endemic areas [41]. Multi-epitope proteins produced in eukaryotic systems of protein expression usually show satisfactory yields [42], and are satisfactorily antigenic.



Fig 4. Indirect anti-ZIKV IgG in-house ELISA. Multi-epitope proteins were evaluated as antigens in an ELISA, using ZIKV+, DENV+, CHIKV+, RF+ patient samples, and negative samples. (A) ZIKV-1 multi-epitope protein as antigen as antigen as antigen multi-epitope protein as antigen and its ROC curve. (B) ZIKV-2 multi-epitope protein as antigen and its ROC curve. (C) ZIKV-3 multi-epitope protein as antigen and its ROC curve. Indexes consider the ratio between absorbance readings and cut-off values. Samples dispersions are represented, and significant differences are highlighted with (*** for p>0.001 or

**** for p>0.0001). ZIKV-/DENV-: negative samples for Zika and dengue; ZIKV+: positive samples for Zika; DENV+: positive samples for dengue; CHIKV+: positive samples for CHIKV; RF+: samples with increased levels of rheumatoid factor.

https://doi.org/10.1371/journal.pntd.0012100.g004

Bacteria-made antigens, are much cheaper to be produced, but may have antigenic problems. Considering the advantages of large-scale production in prokaryotes, we sought to design, produce, and evaluate recombinant proteins that would be promising antigens in a Zika/Dengue differential test.

In the ELISA assays, the multi-epitope proteins showed a satisfactory performance with the first set of sera samples, mainly the ZIKV-1 and ZIKV-2 proteins, where 91.2% and 95% sensitivity were obtained, respectively. In addition, high specificity regarding negative samples was obtained (98.1% and 96.2%, respectively), as well as specificity against DENV+ samples (90.3% and 100%, respectively). In addition, as expected, there was no cross-reactivity with CHIKV + nor with sera from patients with increased FR levels, which is usually considered as a potent interferent in serological tests. Performance assessed with the second set of sera samples corroborated initial results with a similar performance for ZIKV-1 and ZIKV-2 proteins, presenting sensitivity of 88.9% and 97.2%, respectively. In addition, similar results were also observed for their specificity, with 96.9% for both proteins regarding negative samples, as well as 96% and 98.9% specificity regarding DENV+ samples, respectively. The number of samples used in our work was severely limited by sera availability. Due to the difficulties in correctly diagnosing arboviruses during epidemics in affected countries, the number of reliable samples we had access to is not so large. However, at this proof-of-principle stage, we believe the sample panel was sufficient to perform the tests and validations. Further validations of the proposed tests will surely require a larger sample size. The area under the curve (AUC) was 0.97 and 0.95 for ZIKV-1, whilst 0.99 and 0.97 for ZIKV-2, demonstrating a high accuracy, since the AUC represents the overall performance of each antigen, taking into account the sensitivity values of each and their cut-off points. The greater the capability of a test to discriminate between seropositive and seronegative samples, the closer the curve approaches the upper left corner. Thus, the closer the AUC is to 1, the better the test performance [43].

On the other hand, the ZIKV-3 protein showed an overall lower performance regarding sensitivity and specificity on ELISA. One could argue about its larger 3D structure, showing conformational differences and a higher accessible surface, potentially resulting in unspecific protein-protein interactions with other antibodies, such as those from DENV+ samples. This is supported by the binding sites availability and SASA calculations, which could suggest the deviation of antibody interactions to more unspecific fractions of the target protein. This is important as SASA is a decisive factor in protein folding, conformation, and stability studies, including biophysical and thermodynamic properties of the proteins, which would reflect on protein-protein interactions, including the atomic structure and recognition sites of antibodyantigen. Additionally, finding the most bound-like conformation of unliganded protein structures, such as those assessed here, could be related to the presence of both sequences (GPGPGTVNMAEVRSYCYEASIS and GPGPGSVEGELNAILEENGV) intercalations within the ZIKV-3 structure, exposed in its 3D conformation. These could also be associated with unspecific interactions to other antibodies (lower specificity), as aforementioned from the diversified samples used in the tests, as well as hiding (i.e., burying) important residues for specific interactions with ZIKV+ (lower sensitivity). However, no empiric analyses were performed after protein expression, so it cannot be guaranteed that protein folding occurred as predicted after purification.



Fig 5. Indirect anti-ZIKV IgG in-house ELISA, tested at the FAMERP facility, SP, Brazil. (A) ZIKV-1 multiepitope protein as antigen and its respective ROC curve. (B) ZIKV-2 multi-epitope protein as antigen and its respective ROC curve. (C) ZIKV-3 multi-epitope protein as antigen and its respective ROC curve. Indexes consider the ratio between absorbance readings and cut-off values. Samples dispersions are represented, and significant differences are highlighted with (**** for p>0.0001). ZIKV-/DENV-: negative samples for Zika and dengue; ZIKV+: positive samples for Zika; DENV+: positive samples for dengue.

https://doi.org/10.1371/journal.pntd.0012100.g005

In our study, the accessed sera samples came from two different states: from several cities in Minas Gerais state, Brazil, obtained during the Zika outbreak in 2015; and from São Paulo state, Brazil, from a cohort study in the city of São José do Rio Preto. This is important because we could validate the antigenicity of the multiepitope constructs in different epidemiological settings, providing a diversity of samples. Furthermore, the epitopes present in the proteins show high conservation between strains of African and Asian origin. An analysis made with ten strains of African origin and thirty-five strains of Asiatic origin shows that the Epitope 1 (HMCDATMSYECPMLDEGV) has 100% similarity with African strains and 100% with Asiatic strains. The Epitope 2 (HKKGEARRSRRAVTLPSH) has 100% similarity with African strains and only one of the Asian strains showed a difference in one amino acid when compared to the epitope sequence. The Epitope 3 (TVNMAEVRSYCYEASIS) has 88% similarity with the African and Asian strains. The Epitope 4 (TGVFVYNDVEAWRDRYKY) has a valine amino acid in position 5, five strains also have the valine amino acid while five other strains have the isoleucine amino acid, however, they have 100% similarity with the Asiatic strains. The Epitope 5 (SVEGELNAILEENGV) has 100% similarity with African strains and 100% similarity with thirty-one Asian strains and one amino acid difference with the other four strains. The Epitope 6 (WGKSYFVRAAKTNNSFVVDGDTLKECPLKH) has 100% similarity with nine African strains and presented a difference of one amino acid for one strain and 100% similarity with thirty-three Asian strains and presented a difference of one amino acid for two strains. The Epitope 7 (PAVIGTAVKGKEAVH) has 100% similarity with nine African strains and presented a difference of one amino acid for one strain and 100% similarity with thirty-two Asian strains and a difference of one amino acid with three strains (Supplementary material). This high conservation suggests that the multiepitope proteins can be used not only with Brazilian samples but also with samples from other countries.

The developed tests presented superior results compared to available commercial kits, since it was possible to maintain high sensitivity and specificity and, importantly, low cross-reactivity with DENV+ sera. For instance, Low et al. (2021), evaluated two Zika IgG commercial ELISA tests, both of which use the NS1 protein as an antigen, using a panel comprising 278 sera or plasma samples [44]. The Diapro ZIKV IgG test showed an overall specificity of 76.7% and a sensitivity of 84.2%, and when evaluated against DENV+ patients, it showed 8.3% positivity for primary infections and 55.6% for secondary infections. The Euroimmun ZIKV IgG test on the other hand showed an overall specificity of 90%, but a low sensitivity of 52.6%. In addition, when evaluating the test with DENV+ patients, authors found no positivity for primary infections and 22.2% for secondary infections [44]. Morales and coworkers also evaluated the Euroimmun ZIKV IgG test with 907 samples and observed a low sensitivity of 41.4% for acute phase samples, and 72% sensitivity for convalescent samples. As for the negative samples, 9.4% of positivity was obtained, while 33.3% of DENV+ samples reacted positively [45]. Similarly, L'Huillier et al. (2017) also evaluated the Euroimmun test with 223 samples and obtained an overall sensitivity of 23.7% and a specificity of 95.2% [46].

In this work, we highlight the relevance of the use of chimeric proteins as an alternative and potential candidate for differential diagnosis of Zika, especially when considering dengue-positive samples. Additionally, the combination of epitopes derived from several distinct viral proteins with low reactivity among flaviviruses is clearly advantageous when compared to the use of full-length proteins. The successful use of multi-epitope chimeric proteins has been increasingly documented in the literature, and includes chimeric proteins for the diagnosis of *Trypanosoma cruzi* [47], *Burkholderia pseudomallei* [48], *Human T-cell lymphotropic virus* [49,50], *Brucella* [51], *Leishmania braziliensis* [52], among others. These publications also demonstrate the reliability and feasibility of a multi-epitope recombinant approach for diagnosis. Further

developments of the diagnostic tool proposed here will include testing IgM positive samples for Dengue or Zika, as IgM is an important marker for either recent or ongoing infection.

Supporting information

S1 Table. Linear epitopes for B cells found in the literature and tested in serological tests for Zika virus.

(DOCX)

S1 Fig. Alignment of African and Asian strains with the epitopes that compose the multiepitope proteins. (A) Alignment of Epitope 1 with ten African strains and thirty-five Asian strains. (B) Alignment of Epitope 2 with ten African strains and thirty-five Asian strains. (C) Alignment of Epitope 3 with ten African strains and thirty-five Asian strains. (D) Alignment of Epitope 4 with ten African strains and thirty-five Asian strains. (D) Alignment of Epitope 4 with ten African strains and thirty-five Asian strains. (E) Alignment of Epitope 5 with ten African strains and thirty-five Asian strains. (F) Alignment of Epitope 6 with ten African strains and thirty-five Asian strains. (G) Alignment of Epitope 7 with ten African strains and thirty-five Asian strains. (PDF)

Acknowledgments

The authors are very thankful to the Laboratory of Virology and Rickettsioses of the Ezequiel Dias Foundation (FUNED) for providing sera samples. We also thank Edel Figueiredo Barbosa Stancioli for helpful discussions, and Pedro A. Alves for aiding with bioinformatic analyses.

Author Contributions

Conceptualization: Samille Henriques Pereira, Flávio Guimarães da Fonseca.

Data curation: Flávio Guimarães da Fonseca.

Formal analysis: Samille Henriques Pereira, Mateus Sá Magalhães Serafim, Thaís de Fátima Silva Moraes, Nathalia Zini, Jônatas Santos Abrahão, Maurício Lacerda Nogueira, Jordana Grazziela Alves Coelho dos Reis, Flávia Fonseca Bagno, Flávio Guimarães da Fonseca.

Funding acquisition: Flávio Guimarães da Fonseca.

Investigation: Samille Henriques Pereira.

Methodology: Samille Henriques Pereira, Thaís de Fátima Silva Moraes, Nathalia Zini, Flávia Fonseca Bagno.

Project administration: Flávio Guimarães da Fonseca.

Software: Mateus Sá Magalhães Serafim.

Supervision: Flávio Guimarães da Fonseca.

Validation: Nathalia Zini, Flávia Fonseca Bagno.

Writing - original draft: Samille Henriques Pereira, Flávio Guimarães da Fonseca.

Writing – review & editing: Samille Henriques Pereira, Mateus Sá Magalhães Serafim, Thaís de Fátima Silva Moraes, Nathalia Zini, Jônatas Santos Abrahão, Maurício Lacerda Nogueira, Jordana Grazziela Alves Coelho dos Reis, Flávia Fonseca Bagno, Flávio Guimarães da Fonseca.

References

- DICK GWA KITCHEN SF, HADDOW AJ. Zika virus. I. Isolations and serological specificity. Trans R Soc Trop Med Hyg. 1952 Sep; 46(5):509–20.
- Faria NR, do S da S Azevedo R, Kraemer MUG, Souza R, Cunha MS, Hill SC, et al. Zika virus in the Americas: Early epidemiological and genetic findings. Science. 2016 Apr; 352(6283):345–9. https://doi. org/10.1126/science.aaf5036 PMID: 27013429
- Peiter PC, Pereira RDS, Nunes Moreira MC, Nascimento M, Tavares M de FL, Franco V da C, et al. Zika epidemic and microcephaly in Brazil: Challenges for access to health care and promotion in three epidemic areas. PLoS One. 2020; 15(7):e0235010. <u>https://doi.org/10.1371/journal.pone.0235010</u> PMID: 32634152
- 4. Ferraris P, Yssel H, Missé D. Zika virus infection: an update. Microbes Infect. 2019; 21(8–9):353–60. https://doi.org/10.1016/j.micinf.2019.04.005 PMID: 31158508
- Secretária de Vigilância em Saúde e Ambiente—Ministério da Saúde do Brasil. Boletim Epidemiológico 1. Vol. 54. 2023.
- Secretária de Vigilância em Saúde e Ambiente—Ministério da Saúde do Brasil. No Title. Bol Epidemiol 5. 2023;54.
- Center for Disease Control and Prevention (CDC). Zika virus [Internet]. 2019 [cited 2023 Apr 14]. Available from: https://www.cdc.gov/zika/symptoms/symptoms.html.
- Hasan S, Saeed S, Panigrahi R, Choudhary P. Zika Virus: A Global Public Health Menace: A Comprehensive Update. J Int Soc Prev Community Dent. 2019; 9(4):316–27. <u>https://doi.org/10.4103/jispcd.</u> JISPCD_433_18 PMID: 31516865
- Bingham AM, Cone M, Mock V, Heberlein-Larson L, Stanek D, Blackmore C, et al. Comparison of Test Results for Zika Virus RNA in Urine, Serum, and Saliva Specimens from Persons with Travel-Associated Zika Virus Disease—Florida, 2016. MMWR Morb Mortal Wkly Rep. 2016 May; 65(18):475–8. https://doi.org/10.15585/mmwr.mm6518e2 PMID: 27171533
- Land KJ, Boeras DI, Chen X-S, Ramsay AR, Peeling RW. REASSURED diagnostics to inform disease control strategies, strengthen health systems and improve patient outcomes. Nat Microbiol. 2019 Jan; 4 (1):46–54. https://doi.org/10.1038/s41564-018-0295-3 PMID: 30546093
- Musso D, Gubler DJ. Zika Virus. Clin Microbiol Rev. 2016 Jul; 29(3):487–524. https://doi.org/10.1128/ CMR.00072-15 PMID: 27029595
- Wen J, Shresta S. Antigenic cross-reactivity between Zika and dengue viruses: is it time to develop a universal vaccine? Curr Opin Immunol. 2019 Aug; 59:1–8. https://doi.org/10.1016/j.coi.2019.02.001 PMID: 30884384
- Priyamvada L, Quicke KM, Hudson WH, Onlamoon N, Sewatanon J, Edupuganti S, et al. Human antibody responses after dengue virus infection are highly cross-reactive to Zika virus. Proc Natl Acad Sci U S A. 2016 Jul; 113(28):7852–7. https://doi.org/10.1073/pnas.1607931113 PMID: 27354515
- Stettler K, Beltramello M, Espinosa DA, Graham V, Cassotta A, Bianchi S, et al. Specificity, cross-reactivity, and function of antibodies elicited by Zika virus infection. Science. 2016 Aug; 353(6301):823–6. https://doi.org/10.1126/science.aaf8505 PMID: 27417494
- Bagno FF, Figueiredo MM, Villarreal J, Pereira G de C, Godoi LC, da Fonseca FG. Undetected Chikungunya virus co-infections in a Brazilian region presenting hyper-endemic circulation of Dengue and Zika. J Clin Virol Off Publ Pan Am Soc Clin Virol. 2019 Apr; 113:27–30.
- Gutsche I, Coulibaly F, Voss JE, Salmon J, d'Alayer J, Ermonval M, et al. Secreted dengue virus nonstructural protein NS1 is an atypical barrel-shaped high-density lipoprotein. Proc Natl Acad Sci [Internet]. 2011; 108(19):8003–8. Available from: <u>https://www.pnas.org/doi/abs/10.1073/pnas.1017338108</u>. PMID: 21518917
- Kerkhof K, Falconi-Agapito F, Van Esbroeck M, Talledo M, Ariën KK. Reliable Serological Diagnostic Tests for Arboviruses: Feasible or Utopia? Trends Microbiol. 2020 Apr; 28(4):276–92. https://doi.org/ 10.1016/j.tim.2019.11.005 PMID: 31864844
- Lee H-J, Cho Y, Kang HJ, Choi H, Han KR, Chong CK, et al. Identification of peptide based B-cell epitopes in Zika virus NS1. Biochem Biophys Res Commun. 2018 Nov; 505(4):1010–4. <u>https://doi.org/10.1016/j.bbrc.2018.10.024 PMID</u>: 30309651
- 19. Virus Z, Elisa P, Elisa Z, Mishra N, Thakkar R, Ng J, et al. Chapter 10. 2020; 2142:113–21.
- Kam Y-W, Leite JA, Amrun SN, Lum F-M, Yee W-X, Bakar FA, et al. ZIKV-Specific NS1 Epitopes as Serological Markers of Acute Zika Virus Infection. J Infect Dis. 2019 Jun; 220(2):203–12. https://doi.org/ 10.1093/infdis/jiz092 PMID: 30901054

- Kim DTH, Bao DT, Park H, Ngoc NM, Yeo SJ. Development of a novel peptide aptamer-based immunoassay to detect Zika virus in serum and urine. Theranostics. 2018; 8(13):3629–42. <u>https://doi.org/10.7150/thno.25955 PMID: 30026871</u>
- Herrera LR. Immuno informatics Approach in Designing a Novel Vaccine Using Epitopes from All the Structural Proteins of SARS-CoV-2. Biomed Pharmacol J. 2020 Dec 31; 13:1845–62.
- Jumper J, Evans R, Pritzel A, Green T, Figurnov M, Ronneberger O, et al. Highly accurate protein structure prediction with AlphaFold. Nature [Internet]. 2021; 596(7873):583–9. Available from: http://dx.doi. org/10.1038/s41586-021-03819-2. https://doi.org/10.1038/s41586-021-03819-2 PMID: 34265844
- Mirdita M, Schütze K, Moriwaki Y, Heo L, Ovchinnikov S, Steinegger M. ColabFold: making protein folding accessible to all. Nat Methods [Internet]. 2022; 19(6):679–82. Available from: https://doi.org/10. 1038/s41592-022-01488-1. https://doi.org/10.1038/s41592-022-01488-1 PMID: 35637307
- 25. DeLano WL. The PyMOL Molecular Graphics System. Sci LLC. 2006;
- Ngan C-H, Hall DR, Zerbe B, Grove LE, Kozakov D, Vajda S. FTSite: high accuracy detection of ligand binding sites on unbound protein structures. Bioinformatics. 2012 Jan; 28(2):286–7. https://doi.org/10. 1093/bioinformatics/btr651 PMID: 22113084
- Kozakov D, Grove LE, Hall DR, Bohnuud T, Mottarella SE, Luo L, et al. The FTMap family of web servers for determining and characterizing ligand-binding hot spots of proteins. Nat Protoc. 2015 May; 10 (5):733–55. https://doi.org/10.1038/nprot.2015.043 PMID: 25855957
- Fraczkiewicz R, Braun W. Exact and efficient analytical calculation of the accessible surface areas and their gradients for macromolecules. J Comput Chem. 1998; 19(3):319–33.
- Tsai W-Y, Youn HH, Tyson J, Brites C, Tsai J-J, Pedroso C, et al. Use of Urea Wash ELISA to Distinguish Zika and Dengue Virus Infections. Emerg Infect Dis. 2018 Jul; 24(7):1355–9. https://doi.org/10.3201/eid2407.171170 PMID: 29912689
- Sarno M, Sacramento GA, Khouri R, do Rosário MS, Costa F, Archanjo G, et al. Zika Virus Infection and Stillbirths: A Case of Hydrops Fetalis, Hydranencephaly and Fetal Demise. PLoS Negl Trop Dis. 2016 Feb; 10(2):e0004517. https://doi.org/10.1371/journal.pntd.0004517 PMID: 26914330
- Magalhães ICL, Souza PFN, Marques LEC, Girão NM, Araújo FMC, Guedes MIF. New insights into the recombinant proteins and monoclonal antibodies employed to immunodiagnosis and control of Zika virus infection: A review. Int J Biol Macromol. 2022 Mar; 200:139–50. <u>https://doi.org/10.1016/j.ijbiomac.</u> 2021.12.196 PMID: 34998869
- Breitbach ME, Newman CM, Dudley DM, Stewart LM, Aliota MT, Koenig MR, et al. Primary infection with dengue or Zika virus does not affect the severity of heterologous secondary infection in macaques. PLoS Pathog. 2019 Aug; 15(8):e1007766. <u>https://doi.org/10.1371/journal.ppat.1007766</u> PMID: 31369649
- Secretária de Vigilância em Saúde e Ambiente—Ministério da Saúde do Brasil. No Title. Inf Sem 7. 2024;
- Fagbami AH. Zika virus infections in Nigeria: virological and seroepidemiological investigations in Oyo State. J Hyg (Lond). 1979 Oct; 83(2):213–9. <u>https://doi.org/10.1017/s0022172400025997</u> PMID: 489960
- Yan G, Pang L, Cook AR, Ho HJ, Win MS, Khoo AL, et al. Distinguishing Zika and Dengue Viruses through Simple Clinical Assessment, Singapore. Emerg Infect Dis. 2018 Aug; 24(8):1565–8. <u>https://doi.org/10.3201/eid2408.171883 PMID: 30016242</u>
- 36. Pereira SS, Andreata-Santos R, Pereira LR, Soares CP, Félix AC, de Andrade P de MJC, et al. NS1based ELISA test efficiently detects dengue infections without cross-reactivity with Zika virus. Int J Infect Dis IJID Off Publ Int Soc Infect Dis. 2021 Nov; 112:202–4.
- Chibueze EC, Tirado V, Lopes K da S, Balogun OO, Takemoto Y, Swa T, et al. Zika virus infection in pregnancy: a systematic review of disease course and complications. Reprod Health. 2017 Feb; 14 (1):28. https://doi.org/10.1186/s12978-017-0285-6 PMID: 28241773
- Marbán-Castro E, Goncé A, Fumadó V, Romero-Acevedo L, Bardají A. Zika virus infection in pregnant women and their children: A review. Eur J Obstet Gynecol Reprod Biol. 2021 Oct; 265:162–8. https:// doi.org/10.1016/j.ejogrb.2021.07.012 PMID: 34508989
- Charrel RN, Leparc-Goffart I, Pas S, de Lamballerie X, Koopmans M, Reusken C. Background review for diagnostic test development for Zika virus infection. Bull World Health Organ. 2016 Aug; 94(8):574– 584D. https://doi.org/10.2471/BLT.16.171207 PMID: 27516635
- Zamarina T V, Khrapova NP, Pimenova E V, Khanani EI, Viktorov D V, Toporkov A V. [Zika fever immunodiagnostics: overview of test systems.]. Vopr Virusol. 2019; 64(4):150–5. https://doi.org/10.36233/ 0507-4088-2019-64-4-150-155 PMID: 32163680
- 41. Medina FA, Vila F, Premkumar L, Lorenzi O, Paz-Bailey G, Alvarado LI, et al. Capacity of a Multiplex IgM Antibody Capture ELISA to Differentiate Zika and Dengue Virus Infections in Areas of Concurrent

Endemic Transmission. Am J Trop Med Hyg. 2021 Dec; 106(2):585–92. https://doi.org/10.4269/ajtmh. 20-1651 PMID: 34929668

- 42. Lopes-Luz L, Junqueira IC, da Silveira LA, de Melo Pereira BR, da Silva LA, Ribeiro BM, et al. Dengue and Zika virus multi-epitope antigen expression in insect cells. Mol Biol Rep. 2020 Oct; 47(10):7333–40. https://doi.org/10.1007/s11033-020-05772-1 PMID: 32997310
- 43. Hosmer Jr. DW, Lemeshow S, Sturdivant RX. Applied Logistic Regression. 3rd ed. 2013.
- **44.** Low SL, Leo YS, Lai YL, Lam S, Tan HH, Wong JCC, et al. Evaluation of eight commercial Zika virus IgM and IgG serology assays for diagnostics and research. PLoS One. 2021; 16(1):e0244601.
- 45. Morales I, Rosenberger KD, Magalhaes T, Morais CNL, Braga C, Marques ETA, et al. Diagnostic performance of anti-Zika virus IgM, IgAM and IgG ELISAs during co-circulation of Zika, dengue, and chi-kungunya viruses in Brazil and Venezuela. PLoS Negl Trop Dis. 2021 Apr; 15(4):e0009336. https://doi.org/10.1371/journal.pntd.0009336 PMID: 33872309
- 46. L'Huillier AG, Hamid-Allie A, Kristjanson E, Papageorgiou L, Hung S, Wong CF, et al. Evaluation of Euroimmun Anti-Zika Virus IgM and IgG Enzyme-Linked Immunosorbent Assays for Zika Virus Serologic Testing. J Clin Microbiol. 2017 Aug; 55(8):2462–71. <u>https://doi.org/10.1128/JCM.00442-17</u> PMID: 28566316
- Freitas NEM, Santos EF, Leony LM, Silva ÂAO, Daltro RT, Vasconcelos L de CM, et al. Double-antigen sandwich ELISA based on chimeric antigens for detection of antibodies to Trypanosoma cruzi in human sera. PLoS Negl Trop Dis. 2022 Mar; 16(3):e0010290. <u>https://doi.org/10.1371/journal.pntd.0010290</u> PMID: 35275913
- Wajanarogana S, Taylor WRJ, Kritsiriwuthinan K. Enhanced serodiagnosis of melioidosis by indirect ELISA using the chimeric protein rGroEL-FLAG300 as an antigen. BMC Infect Dis. 2022 Apr; 22 (1):387. https://doi.org/10.1186/s12879-022-07369-4 PMID: 35439967
- Mosadeghi P, Heydari-Zarnagh H. Development and Evaluation of a Novel ELISA for Detection of Antibodies against HTLV-I Using Chimeric Peptides. Iran J Allergy Asthma Immunol. 2018 Apr; 17(2):144– 50. PMID: 29757587
- 50. Franco G de M, da Rocha AS, Cox LJ, Silva DS de O Daian E, da Silveira ESantos DMMartins ML, et al. Multi-Epitope Protein as a Tool of Serological Diagnostic Development for HTLV-1 and HTLV-2 Infections. Front public Heal. 2022; 10:884701. <u>https://doi.org/10.3389/fpubh.2022.884701</u> PMID: 35677763
- Bulashev AK, Ingirbay BK, Mukantayev KN, Syzdykova AS. Evaluation of chimeric proteins for serological diagnosis of brucellosis in cattle. Vet world. 2021 Aug; 14(8):2187–96. https://doi.org/10.14202/ vetworld.2021.2187-2196 PMID: 34566338
- 52. Garcia GC, Carvalho AMRS, Duarte MC, Silva MFCE, Medeiros FAC, Coelho EAF, et al. Development of a chimeric protein based on a proteomic approach for the serological diagnosis of human tegumentary leishmaniasis. Appl Microbiol Biotechnol. 2021 Sep; 105(18):6805–17. https://doi.org/10.1007/ s00253-021-11518-1 PMID: 34432132